\documentclass[12pt]{article}

\usepackage{hyperref}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{amsopn}
\usepackage{amsthm}
\usepackage[all]{xy}

% ****** Set environment for paragraphs ***********

\theoremstyle{plain}
\newtheorem{thm}{Theorem}[section]
\newtheorem{lem}[thm]{Lemma}
\newtheorem{cor}{Corollary}
\newtheorem{prop}[thm]{Proposition}
\newtheorem*{mainthm}{Theorem}

\theoremstyle{definition}
\newtheorem{ntt}[thm]{}
\newtheorem{ex}[thm]{Example}
\newtheorem{rem}[thm]{Remark}
\newtheorem{dfn}[thm]{Definition}

% ****** Notations **************

%\newcommand{\af}{\mathbb{A}}   % affine line
\newcommand{\pl}{\mathbb{P}}   % projective line
%\newcommand{\gm}{{\mathbb{G}_m}} % multiplicative group scheme
\newcommand{\zz}{\mathbb{Z}}   % integers
%\newcommand{\hh}{\mathbb{H}}   % hyperbolic plane
%\newcommand{\oo}{\mathcal{O}}  % structure sheaf
\newcommand{\A}{\mathrm{A}} % An
%\newcommand{\B}{\mathrm{B}} % Bn
\newcommand{\C}{\mathrm{C}} %Cn
\newcommand{\D}{\mathrm{D}} %Dn
\newcommand{\E}{\mathrm{E}} %En
\newcommand{\F}{\mathrm{F}_4} %F4
\newcommand{\G}{\mathrm{G}_2}  % G2
\newcommand{\HH}{\mathrm{H}^1} % H^1
%\newcommand{\w}{\bar{\omega}} % weight
\newcommand{\id}{\mathrm{id}}       % identity
\newcommand{\pr}{\mathrm{pr}}       % projection
%\newcommand{\can}{\mathrm{can}}     % canonical map
\newcommand{\res}{\mathrm{res}}     % restriction
\newcommand{\op}{\mathrm{op}}     % opposite


% ******* Categories ********

%\newcommand{\Var}{\mathcal{V}ar} % projective varieties
\newcommand{\M}{\mathcal{M}}     % Chow motives
%\newcommand{\J}{\mathcal{J}} % tautological bundles
%\newcommand{\E}{\mathrm{E}} %
%\newcommand{\F}{\mathrm{F}} %
%\newcommand{\V}{\mathcal{V}} % tautological bundles
%\newcommand{\OO}{\mathcal{O}} % structre sheaf
\newcommand{\OP}{\mathbb{OP}^2} % Cayley plane
%\newcommand{\Cor}{\mathcal{C}or} % Correspondences
%\newcommand{\zab}{\mathbb{Z}\text{-}\mathcal{A}b} % Z-Abelian groups
%\newcommand{\Sb}{\mathcal{H}} % fake Severi-Brauer motive
%\newcommand{\De}{\mathit{\Delta}} % partition
%\newcommand{\Seg}{\mathrm{Seg}} % Segre embedding
\newcommand{\X}{X} % E7/P7
\newcommand{\B}{\mathcal B} % Brown algebra

% ****** arrows *************

\newcommand{\hra}{\hookrightarrow}     % hook right arrow
\newcommand{\lra}{\longrightarrow}     % long right arrow
\newcommand{\xra}[1]{\xrightarrow{#1}} % right arrow with parameter

% ******* Math Operators - Functors ***********

\DeclareMathOperator{\Spec}{\mathrm{Spec}}  % Spectrum
\DeclareMathOperator{\Spin}{\mathrm{Spin}}  % Spin
\DeclareMathOperator{\Pic}{\mathrm{Pic}}    % Picard group
\DeclareMathOperator{\Br}{\mathrm{Br}}     % Brauer group
\DeclareMathOperator{\Gal}{\mathrm{Gal}}   % Galois group
\DeclareMathOperator{\CH}{\mathrm{CH}}      % Chow group
\DeclareMathOperator{\Ch}{\mathrm{Ch}}
\DeclareMathOperator{\CHO}{\widetilde{\mathrm{CH}}_0}
\DeclareMathOperator{\Aut}{\mathrm{Aut}}    % Automorphisms
\DeclareMathOperator{\Hom}{\mathrm{Hom}}    % Homomorphisms
\DeclareMathOperator{\Sim}{\mathrm{Sim}}    % Similitudes
\DeclareMathOperator{\Mor}{\mathrm{Mor}}    % Morphisms
\DeclareMathOperator{\RatCor}{\mathrm{RatCor}}    
\DeclareMathOperator{\End}{\mathrm{End}}    % Endomorphisms
\DeclareMathOperator{\Ker}{\mathrm{Ker}}    % Kernel
\DeclareMathOperator{\im}{\mathrm{Im}}      % image
\DeclareMathOperator{\SB}{\mathrm{SB}}      % Severi-Brauer variety
\DeclareMathOperator{\K}{\mathrm{K}}        % K-functor
\DeclareMathOperator{\PGL}{\mathrm{PGL}}      % Adjoint group of inner type $A_n$
\DeclareMathOperator{\GO}{\mathrm{GO}}  % GO
\DeclareMathOperator{\Gr}{\mathrm{Gr}} % Grassmann
\DeclareMathOperator{\Fr}{\mathrm{Fr}} % Fraction field
\DeclareMathOperator{\rdim}{\mathrm{rdim}} % reduced dimension
\DeclareMathOperator{\rk}{\mathrm{rk}} % rank
\DeclareMathOperator{\tr}{\mathrm{tr}} % trace
\DeclareMathOperator{\ind}{\mathrm{ind}} % index
\DeclareMathOperator{\sep}{\mathrm{sep}} % separable closure
\DeclareMathOperator{\ad}{\mathrm{ad}} % separable closure
\DeclareMathOperator{\Stab}{\mathrm{Stab}}  % Stabilizer
\DeclareMathOperator{\car}{\mathrm{char}}  % characteristic
\DeclareMathOperator{\diag}{\mathrm{diag}} % diagonal matrix
\DeclareMathOperator{\cd}{\mathrm{cd}}  % canonical dimension

\title{Canonical $p$-dimensions of algebraic groups and
degrees of basic polynomial invariants}

\author{K.~Zainoulline}


%%%%%%%%%%%%%%%%%%%% DOCUMENT %%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{document}
\begin{thebibliography}{99}



\bibitem{BR} Berhuy, G., Reichstein, Z. On the notion of canonical dimension
for algebraic groups. Advances in Math., in press, doi: 10.1016/j.aim.2004.12.004.

\bibitem{Kc} Kac, V. Torsion in cohomology of compact Lie groups and
Chow rings of reductive algebraic groups. Invent. Math. 80 (1985), 69--79.

\bibitem{Ka} Karpenko, N. Canonical dimension of orthogonal group. Transform. Groups 10 (2005), no.~2, 211--215.

\bibitem{KM} Karpenko, N., Merkurjev, A. Canonical $p$-dimension of algebraic groups.
Advances in Math., in press, doi: 10.1016/j.aim.2005.07.013.

\bibitem{St} Stanley, R. Hilbert functions of graded algebras. Advances in Math. 28 (1978), 57--83.

\end{thebibliography}


\maketitle

\begin{abstract}
In the present notes we provide a new uniform way to compute 
a canonical $p$-dimension of a split algebraic group $G$ for
a torsion prime $p$
using degrees of basic polynomial invariants described by V.~Kac.
As an application, we compute the canonical $p$-dimensions
for all split simple algebraic groups.
\end{abstract}

The notion of a canonical dimension of an algebraic structure
was introduced by Berhuy and Reichstein \cite{BR}.
For a split algebraic group $G$ and its torsion prime $p$ 
the canonical $p$-dimension of $G$ was studied by 
Karpenko and Merkurjev in \cite{Ka} and \cite{KM}.
In particular, this invariant was shown to be related with
the size of the image of the characteristic map 
\begin{equation}
\phi_G: S^*(\hat{T}) \to \CH^*(X),
\end{equation}
where $\hat{T}$ is the character group of a maximal split torus $T$,
$X=G/B$ is the variety of complete flags and $S^*$ stands for the
 symmetric algebra.
Namely, one has the following formula for 
the canonical $p$-dimension of a group $G$
\begin{equation}
\cd_p(G)=\min \{i \mid \overline{\Ch_i}(X)\},
\end{equation}
where $\overline{\Ch_i}(X)$ stands for the image of $\phi_G$ in the Chow
ring $\Ch_i(X)=\CH_i(X;\zz/p\zz)$ with $\zz/p\zz$-coefficients.
Using this remarkable fact together with the explicit description
of the image $R_p=\overline{\Ch}(X)$
Karpenko and Merkurjev computed the canonical $p$-dimensions
for all classical algebraic groups $G$.


The goal of the present notes is to relate
the work by V.~Kac \cite{Kc}
devoted to the study of the $p$-torsion of the Chow ring of an algebraic group $G$
with the canonical $p$-dimensions of $G$.
In particular we provide a different and uniform approach
of computing $cd_p(G)$.

\begin{mainthm} Let $G$ be a split simple algebraic group of rank $n$
and $p$ be a torsion prime. 
Then
$$
\cd_p(G)=N + n - (d_{1,p}+ d_{2,p}+\ldots+d_{n,p}),
$$
where $N$ stands for the number of positive roots of $G$ and 
integers $d_{1,p},\ldots,d_{1,n}$ are 
the degrees of basic polynomial invariants modulo $p$.
\end{mainthm}

\begin{proof} Consider the characteristic map (1) modulo $p$
\begin{equation}
(\phi_G)_p: S^*(\hat{T})\otimes_\zz \zz/p\zz \to \Ch^*(X)
\end{equation}
According to \cite[Theorem 1]{Kc}, 
the kernel of this map $I_p$ is generated by a regular sequence
of $n$ homogeneous polynomials of degrees $d_{1,p},\ldots,d_{n,p}$.

Recall that a Poincare polynomial $P(A,t)$ for a graded module $M^*$ 
over a field $k$ is defined to be
$\sum_i \dim_k M^i \cdot t^i$ (see \cite{St}).
Hence, the Poincare polynomial for the $\zz/p\zz$-module 
$R_p$ is equal to
\begin{equation}
P(R_p,t)= \prod_{i=0}^n \frac{1-t^{d_{i,p}}}{1-t}.
\end{equation}
Indeed, we identify $R_p$ with the quotient of the polynomial ring
in $n$ variables $\zz/p\zz[\omega_1,\ldots,\omega_n]$ modulo the ideal $I_p$
(here $\omega_i$ are the fundamental weights).
The formula (4) then follows immediately by \cite[Cor.~3.3]{St}.

According to (2) the canonical $p$-dimension is equal to the difference
$\dim(X)- \deg P(R_p,t)$.
\end{proof}

\begin{cor} We obtain the following values for the canonical $p$-dimensions
of groups of types $F_4$, $E_6$, $E_7$ and $E_8$ (in the list below
$G^{sc}$ and $G^{ad}$ denote the simply-connected and adjoint forms of $G$).

\begin{tabular}{ccc}
$\cd_2 F_4 = 3$, & $\cd_3 F_4 = 8$ & \\
$\cd_2 E_6 = 3$, & $\cd_3 E_6^{sc} = 8$, & $\cd_3 E_6^{ad} = 16$ \\
$\cd_2 E_7^{sc} = 17$, & $\cd_2 E_7^{ad} = 18$ & $\cd_3 E_7 = 8$ \\
$\cd_2 E_8 = 60$, & $\cd_3 E_8= 28$ & $\cd_5 E_8= 24$ 
\end{tabular}

\end{cor}

\begin{proof} Follows from the list of degrees of basic polynomial invariants
provided in \cite[Table 2]{Kc}.
\end{proof}

\begin{cor}
There is the following relation between
the canonical $p$-dimension of $G$ and the $p$-torsion part of the Chow group of $G$
$$
\cd_p(G)= \max \{i \mid \Ch^i(G)\neq 0 \}
$$
\end{cor}

\begin{proof} It is known that $\Ch(G)=\Ch(X)/J_p$,
where $J_p$ is the ideal generated by the non-constant part of $R_p$.
Since $\Ch(X)$ is a free $R_p$-module, we obtain
$P(\Ch(X)/J_p,t)=P(\Ch(X),t)/P(R_p,t)$.
\end{proof}

\begin{cor}
Let $d_1,d_2,\ldots,d_n$ be the degrees of ``usual'' basic polynomial
invariants of $G$, i.e., with $\mathbb{Q}$-coefficients. Then the canonical
$p$-dimension of $G$ is equal to
$$
\cd_p(G) = \sum_{i=0}^n d_i'\cdot (p^{k_i}-1),
$$
where the integers $d_i'$ and $p^{k_i}$ are the factors of the decompositions
$d_i=d_i'\cdot p^{k_i}$, $p \nmid d_i'$. Observe that the degrees $d_i$ for
which $k_i>0$ form the set of $p$-exceptional degrees introduced in \cite{Kc}.
\end{cor}

\begin{proof} Follows by the isomorphism of \cite[Theorem 3,(ii)]{Kc}.
\end{proof}

{\it Acknowledgements.} 
I am grateful to Burt Totaro for mentioning to me
the paper \cite{Kc}.

\bibliographystyle{chicago}


\end{document}
